Author’s response to reviews

Title: Hand disinfection in a neonatal intensive care unit: Continuous electronic monitoring over a one-year period.

Authors:

Onno Helder (o.helder@erasusmc.nl)
Johannes van Goudoever (h.vangoudoever@vumc.nl)
Wim Hop (w.hop@erasusmc.nl)
Johannes Brug (j.brug@vumc.nl)
René Kornelisse (r.kornelisse@erasusmc.nl)

Version: 2 Date: 15 February 2012

Author’s response to reviews:

We sent you the requested Title page, manuscript including the abstract end included the name of the ethnics committee.